Author's response to reviews

Title: Use of concept mapping to characterize relationships among implementation strategies and assess their feasibility and importance: Results from the Expert Recommendations for Implementing Change (ERIC) study

Authors:

Thomas J Waltz (twaltz1@emich.edu)
Byron J Powell (byronp@upenn.edu)
Monica M Matthieu (mmatthie@slu.edu)
Laura J Damschroder (laura.damschroder@va.gov)
Matthew J Chinman (chinnman@rand.org)
Jeffrey L Smith (jeffrey.smith6@va.gov)
Enola K Proctor (ekp@wustl.edu)
JoAnn E Kirchner (joann.kirchner@va.gov)

Version: 2
Date: 9 April 2015

Author's response to reviews:

The following sections have been added to the end of the manuscript: contributors, abbreviations, competing interests, authors’ contributions, acknowledgements.